Author’s response to reviews

Title: Identification of gene mutations in patients with primary periodic paralysis using targeted next-generation sequencing

Authors:

Sushan Luo (susanro_36@hotmail.com)
Minjie Xu (11210700016@fudan.edu.cn)
Jian Sun (sunjian19927@yeah.net)
Kai Qiao (kateqiao@hotmail.com)
Jie Song (sdsj1990@163.com)
Shuang Cai (shuang4601441@126.com)
Wenhua Zhu (jessiblue@yeah.net)
Lei Zhou (zhoulei0156022@hotmail.com)
Jianying Xi (xijiangying@fudan.edu.cn)
Jiahong Lu (lujiahong@fudan.edu.cn)
Xiaohua Ni (nixiaohua@sippr.org.cn)
Tonghai Dou (tonghaidou@fudan.edu.cn)
Chongbo Zhao (zhao_chongbo@fudan.edu.cn)

Version: 4 Date: 26 Apr 2019

Author’s response to reviews:

Dear Editorial board,

Thanks so much for detailed comments for the format. We addressed the questions listed in this revision. Please do not hesitate to contact us if we need to revise anything else.

Best wishes,

Chongbo Zhao
Response for the Reviewer reports:

1. In the Consent to publish subsection of the Declarations, please declare whether consent was obtained in written or verbal form. Please note that in accordance with BMC submission guidelines, written informed consent must be obtained for the publication of identifiable material. Therefore, if written informed consent was not obtained, please take the opportunity to obtain it now from participants or remove potential identifiers from the dataset. Alternatively, you may remove the data set and make your data available upon request from the authors.

   Additionally, please declare whether consent for publication was provided by parents and/or caregivers on behalf of underage participants.

   Response: We adjust these issues accordingly.

2. On the title page please include the institutional email addresses of all authors.

   Additionally, please declare only one corresponding author.

   Response: Prof. Chongbo Zhao is the only corresponding author.

3. In your Funding section, please also state the role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript.

   Response: We adjust these issues accordingly.

4. In the Authors’ contributions subsection, please refer to authors using their full initials rather than their full names. If two authors have the same initials, please distinguish between them by designating them with numbers 1 and 2. For example: JS1 and JS2. JS1 would correspond to the author furthest up on the author list.

   Response: We adjust these issues accordingly.

5. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

   An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. Authors are expected to fulfil the criteria below (adapted from McNutt et al., Proceedings of the National Academy of Sciences, Feb 2018, 201715374; DOI: 10.1073/pnas.1715374115; licensed under CC BY 4.0):
Each author is expected to have made substantial contributions to the conception OR design of the work; OR the acquisition, analysis, OR interpretation of data; OR the creation of new software used in the work; OR have drafted the work or substantively revised it

AND to have approved the submitted version (and any substantially modified version that involves the author's contribution to the study);

AND to have agreed both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.

Acquisition of funding, collection of data or general supervision of the research group, alone, does not usually justify authorship.

Anyone listed as an author must be included in this section. If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found at this web address https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5. All instructions can be found on the form. Please return the form within 14 days with all author signatures.

Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.

Response: We revised the authors’ contribution section and acknowledgement accordingly.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We upload the renewed manuscript as a clean version without any highlighted tracks.

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandreviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.
Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements, this may cause a delay while this is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

In accordance with BioMed Central editorial policies and formatting guidelines, all manuscript submissions to BMC Neurology must contain a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal's Submission Guidelines web page for information regarding the criteria for each sub-section (https://bmcneurol.biomedcentral.com/).

Where a mandatory Declarations section is not relevant to your study design or article type, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must confirm this under this sub-heading and also provide their reasons. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page (see links below).

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements

Further information about our editorial policies can be found at the following links:

Ethical approval and consent:

http://www.biomedcentral.com/about/editorialpolicies#Ethics
Availability of data and materials section:

http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials

Recipients of this email are registered users within the Editorial Manager database for this journal. We will keep your information on file to use in the process of submitting, evaluating and publishing a manuscript. For more information on how we use your personal details please see our privacy policy at https://www.springernature.com/production-privacy-policy. If you no longer wish to receive messages from this journal or you have questions regarding database management, please contact the Publication Office at the link below.

____________________________________________________________________________________

In compliance with data protection regulations, you may request that we remove your personal registration details at any time. (Use the following URL: http://nurl.edmgr.com/login.asp?a=r) Please contact the publication office if you have any questions.